Author’s response to reviews

Title: Giving birth in Switzerland: a qualitative study exploring migrant women’s experiences during pregnancy and childbirth in Geneva and Zurich using focus groups

Authors:

Jana Sami (samijana04@gmail.com)
Katharina Quack Lötscher (Katharina.QuackLoetscher@usz.ch)
Isabelle Eperon (isabelle.eperon@hcuge.ch)
Lea Gonik (Lea.Gonik@hcuge.ch)
Begona Martinez de Tejada (begona.martinezdetejada@hcuge.ch)
Manuella Epiney (Manuella.Epiney@hcuge.ch)
Nicole Schmidt (nicole.ch.schmidt@gmail.com; nicole.schmidt@hcuge.ch)

Version: 2 Date: 30 Jun 2019

Author’s response to reviews:

Dear Editor in Charge

Dear Reviewers

Thank you for revising our manuscript. Please find attached a revised version (with track changes) and a clean version. We have included the recommendation of table 3 and the enumerations, which have been previously in the conclusion section, in the discussion part. In this process the former table 3 was eliminated as the recommendations were included in the discussion. The conclusion was shortened and formulated more precisely.

We hope that you find the revised manuscript suitable for publication. Please do not hesitate to contact us.

Kind regards

Nicole Schmidt